Author's response to reviews

Title: What stops children with a chronic illness accessing health care: a mixed methods study in children with CFS/ME

Authors:

Carly M Webb (m0702395@sgul.ac.uk)
Simon M Collin (simon.collin@bristol.ac.uk)
Toity Deave (toity.deave@uwe.ac.uk)
Andrew Haig-Ferguson (andrew.haig-ferguson@nhs.net)
Amy Spatz (aspatz@sgul.ac.uk)
Esther Crawley (esther.crawley@bristol.ac.uk)

Version: 2 Date: 26 May 2011

Author's response to reviews: see over
Dear Dr Patel,

We have pleasure in submitting our revision of the paper ‘What stops children with a chronic illness accessing health care: a mixed methods study in children with CFS/ME’

We have revised the following points within the qualitative methods:

- Eligibility criteria
- Recruitment process by clinical team
- The process of composing the content of the interviews in collaboration with the Association of Young people with ME.
- Qualitative analysis
  - Process of analysis
  - Who conducted analysis
  - Data validation and rigour

The following references were added:


Thank you for considering our paper.

Dr Jigisha Patel
Series Editor
BMC Family Practice

A joint initiative between University of Bristol and University of the West of England, Bristol

www.bris.ac.uk/ccah
Dr Esther Crawley

A joint initiative between University of Bristol and University of the West of England, Bristol

www.bris.ac.uk/ccah